Author's response to reviews

Title: Secondhand smoke exposure toxicity accelerates age-related cardiac disease in old hamsters

Authors:

Jia-Ping Wu (affymax0823@hotmail.com.tw)
Chih-Yang Huang (cyhuang@mail.cmu.edu.tw)
Cheng-Hong Hsieh (hyweuan@asia.edu.tw)
Tsung-Jung Ho (tjho@mail.cmu.edu.tw)
Wei-Wen Kuo (wwkuo@mail.cmu.edu.tw)
Yu-Lan Yeh (1867@cch.org.tw)
Chien-Chung Lin (linchienchung@yahoo.com.tw)
Chia-Hua Kuo (kuochiahua@gmail.com)
Fuu-Jen Tsai (d0704@mail.cmu.edu.tw)
Chang-Hai Tsai (chchai@mail.cmu.edu.tw)

Version: 3  Date: 3 November 2014

Author's response to reviews:

Dear Editor:

It is this property of the hypertrophied heart which the heart muscle enables it to maintain the circulation when increased work is required of it. The hypertrophied heart must also have this power of accommodating itself to altered conditions, however, the normal heart possesses a reserve force. Once the loss of which cause of heart failure. In this study, there would be evidenced of a failing heart. A left ventricular hypertrophy should place an predominant role resulted from hyperplasia to decompensatory hypertrophy in the systemic circulation. Here thank you accept our manuscript. The revised concerns as following revision cover letter: